Author's response to reviews

Title: Prognostic relevance of Wnt-inhibitory factor-1 (WIF1) and Dickkopf-3 (DKK3) promoter methylation in human breast cancer

Authors:

Jürgen Veeck (juergen.veeck@rwth-aachen.de)
Peter J Wild (peter.wild@cell.biol.ethz.ch)
Thomas Fuchs (thomas.fuchs@inf.ethz.ch)
Peter J Schüffler (peter.schueffler@inf.ethz.ch)
Arndt Hartmann (arndt.hartmann@uk-erlangen.de)
Ruth Knüchel (rknuechel-clarke@ukaachen.de)
Edgar Dahl (edahl@ukaachen.de)

Version: 3 Date: 25 June 2009

Author's response to reviews:

Dear BMC Cancer Production Team,

following your request we have made the following changes to the manuscript:

1. We have changed the authors listing, which now appears exactly the same in the manuscript and in the online submission form
2. Table 4 has been referenced as Table 4a and Table 4b. Accordingly, Table cites have been actualized.
3. When citing figures, all cites now read as "Figure" instead of "Fig".
4. Reference [49] has been eliminated from the manuscript. Reference [50] has been included in the manuscript text and then became reference [27]. Any subsequent reference has been actualized accordingly.
5. Any underlining in the manuscript has been eliminated.
6. The affiliation presented as "Present address" is now listed as "Current address".
7. Typographical errors have been eliminated and the manuscript has been proof-read for spelling consistency.

We hope to have fulfilled any requested formatting changes.

On behalf of the corresponding author,

best regards,

Jürgen Veeck